Author's response to reviews

Title: Cost-Effectiveness of Monitoring Glaucoma Patients in Shared Care: an Economic Evaluation alongside a Randomized Controlled Trial

Authors:

Kim M Holtzer-Goor (goor@bmg.eur.nl)
Esther van Sprundel (evansprundel@home.nl)
Hans G Lemij (hlemij@me.com)
Thomas Plochg (t.plochg@amc.uva.nl)
Niek S Klazinga (N.S.Klazinga@amc.uva.nl)
Marc A Koopmanschap (koopmanschap@bmg.eur.nl)

Version: 4 Date: 26 October 2010

Author's response to reviews: see over
Office
Mr Graham Mowatt

October 26th, 2010

Submission third revision of manuscript (MS: 9206910473385012)

Dear Mr Graham Mowatt,

Please find enclosed the revised version of our manuscript entitled “Cost-Effectiveness of Monitoring Stable Glaucoma Patients in Shared Care: an Economic Evaluation alongside a Randomized Controlled Trial” (MS: 9206910473385012). Thank you for the opportunity to clarify the method of randomisation in this third revision. The clarification has been highlighted in green.

Yours sincerely,
Kim Holtzer-Goor

Erasmus MC, Institute for Medical Technology Assessment
PO Box 1738
3000 DR Rotterdam
The Netherlands
Telephone +31 10 4088539
Fax +31 10 4089081
Email: goor@bmg.eur.nl